





FULL/LONG TITLE OF THE STUDY

Improving triadic decision-making involving people with dementia

SHORT STUDY TITLE / ACRONYM

Making decisions whilst living with memory problems

PROTOCOL VERSION NUMBER AND DATE

Protocol Version 3.0 20 June 2023

RESEARCH REFERENCE NUMBERS

IRAS Number:	321260
SPONSORS Number:	STH 22002
FUNDERS Number:	NIHR 150756
REC Reference	23/WA/0147

VERSION CONTROL

Version	Date	Amendments
Version 1.0	24 March 2023	Final version of protocol.
Version 2.0	30 May 2023	Added end of study definition (Section 5).
Version 3.0	20 June 2023	Full title changed to match the NIHR funding title, version control table added.







KEY STUDY CONTACTS

Chief Investigator	Dr Traci Walker Senior Lecturer Health Sciences School, Division of Human Communication Sciences The University of Sheffield 362 Mushroom Lane Sheffield S10 2TS
Sponsor	Sheffield Teaching Hospital NHS Foundation Trust
Funder(s)	NIHR Health and Social Care Delivery Research (HSDR) Programme

STUDY SUMMARY

Abbreviations used in this document	
ACE	Addenbrooke's Cognitive Examination
CA	Conversation Analysis
HCP	Healthcare professional
PIS	Participant Information Sheet
PPI	Patient and Public Involvement
PwND	Person/people with neurodegenerative cognitive disorder (eg., dementia)
RA	Research Associate
RDaSH	Rotherham, Doncaster and South Humber NHS Foundation Trust
SDM	Shared decision-making
STH	Sheffield Teaching Hospitals NHS Foundation Trust

KEY WORDS: conversation analysis; dementia; multiparty interaction; shared decisionmaking; memory

Study Title	Improving triadic decision-making involving people with dementia
2	







Sheffield Teaching Hospitals NHS Foundation Trust

Internal ref. no. (or short title)	Making decisions whilst living with memory problems
Study Design	Qualitative research
Study Participants	People living with neurodegenerative cognitive disorders
Planned Size of Sample (if applicable)	100 dyads across 2 sites x 1 interaction; 10 co-researcher participant dyads working alongside the research team to collect videorecorded data for up to 2 years
Planned Study Period	4 years
Research Questions	1) What communication strategies are healthcare professionals and carers using to involve people with neurodegenerative cognitive disorders in decision making? 2) How do people with neurodegenerative cognitive disorders respond to these in the consultation?
Research aims and objectives	 There are 4 key research aims: 1. To contribute to the evidence base about how decision making takes place in triadic environments 2. To compare findings on decision making in triadic medical encounters involving patients with and without neurodegenerative cognitive disorders 3. To identify the most effective and appropriate strategies for involving people with neurodegenerative cognitive disorders (PwND) in decisions about their medical care, especially as the disease progresses 4. To disseminate these findings to healthcare professionals (HCP), PwND and carers in an effort to increase patient involvement in decision-making about their care And 6 study objectives: 1. Provide a detailed analysis of the communicative strategies used in triadic consultations and compare those involving patients with or without neurodegenerative cognitive disorders (Aim 1) 2. Identify how these strategies can support or limit shared decision-making for PwND (Aim 1, 2) 3. Use both cross-sectional and a novel longitudinal PwND/carer co-researcher design to describe how the







	meaning of shared decision-making for PwND evolves and communicative strategies used change as the disease progresses (Aim 2, 3)
4.	Test the hypothesis that living with a
	neurodegenerative cognitive disorder is a
	'transformative experience', and the authentic values
	of PwND change as the disease progresses (Aim 3)
5.	Construct educational resources and decision support
	tools to enable shared decision making for PwND
	(Aim 4)
6.	Disseminate results to PwND and all involved in
	working with and caring for them (Aim 4)



STUDY FLOW CHART







ROLES AND RESPONSIBILITIES OF STUDY MANAGEMENT COMMITTEES & GROUPS

PPI

We have 4 PPI co-applicants who have been/will be involved in assuring that the study is relevant and that our communication with research participants and any documents intended for dissemination of the findings to service users are understandable and accessible to potential participants. Our PPI co-applicants will help to design documents such as the PIS and consent forms, and help us to identify innovative ways of approaching potential participants to ensure optimal understanding of our research plans and truly informed consent. These co-applicants will also be involved in the dissemination phase to ensure that the findings get to the appropriate user groups/stakeholders and are communicated in ways that are acceptable, accessible and meaningful to lay audiences.

The inclusion of experts by experience as co-researcher participants in the third part of our study will provide for participant and public involvement and offer additional oversight of the data collection process. We will also be meeting quarterly with our PPI co-applicants to report on data collection and analysis.

In the early design stages of the project, we consulted with the charity DonMentia (which is comprised of people with dementia and their partners). We have been invited to regularly attend their meetings and present updates on the research on a regular basis. This group is well-versed in providing feedback to researchers, and assisted us in the drafting of the research proposal. We also received valuable feedback (eg., to use words like 'choices' instead of 'wishes') from the South Yorkshire Dementia Research Advisory Group (DRAiSY) who were particularly enthusiastic about the longitudinal arm of the study (including the element of video recording).

Study Steering Group

We have set up a study steering committee with 5 independent members (of six members in total), one of whom is a PPI member recruited through the NIHR's People in Research website. This project uses exclusively qualitative methodology, similar to an approach which has been used in previous studies led by one of the co-PIs (Prof Markus Reuber) and funded by the NIHR. The Steering Group will also serve as our internal Data Monitoring and Ethics Committee (in addition to all aspects of the study being subject to external inspection and oversight by the sponsor and regulatory bodies). A separate Data Monitoring and Ethics Committee is not necessary as our study is purely observational and as the kind of data we will be collecting and analysing is not intrusive. None of the study procedures put participants at any significant risk.

STUDY PROTOCOL: Improving triadic decision-making involving people with dementia

Short lay summary:

Most care decisions involving people with neurodegenerative cognitive disorders (PwND) are made in interactions involving three parties: the PwND, their carer, and a healthcare practitioner (HCP). This study explores two gaps in knowledge: how decisions are made when three parties are involved, and how decisions are made when one of those parties has cognitive difficulties. The study aims to describe how best to involve PwND in decision making as much as possible, for as long as possible.







We focus on the following questions: What communication strategies are HCPs and carers using to involve PwND in decision making? How do PwND respond to these? Our study will apply Conversation Analysis (CA) to existing and new datasets, in the form of 3 workpackages.

WP1: Analysis of extant recordings of triadic interactions between neurologists, patients and accompanying persons from previous studies collected in outpatient clinics in Sheffield.

WP2: Collect and analyse a new cross-sectional database of 100 new recordings of clinical encounters between patients who have or subsequently receive a diagnosis of neurodegenerative cognitive disorder, their carers, and HCP.

WP3: Recruit 10 PwND and their carers from WP2 as participant-researchers, to record their medical appointments (eg., with dementia nurses, occupational therapists, GPs) for up to two years. This will allow us to compare interactions with different HCPs and analyse both changes in PwNDs' contributions and in the role of accompanying persons over time. The team will engage both PwND and their carers in deciding what encounters to record, as well as explore their perspectives on when and how they (both) were involved successfully in shared decision-making.

Our outputs will include online training packages, short videos and leaflets. Those for carers will help them support PwND in decision-making, and those for HCP will help ensure that PwND are given every chance to express their choices when making decisions.

1 BACKGROUND

People should be offered the chance to participate in decisions about their health, and people with neurodegenerative cognitive disorders (PwND) have the same rights to autonomy and self-determination as anyone else [1]. However, while the NHS constitution and GMC direct clinicians to ensure the participation of PwND in decision making [2,3], little advice is provided on how this can best be achieved when the patient's capacity is reduced. Additionally, PwND typically bring someone else along to medical appointments [4], and some of the decisions that need to be made directly involve third parties (such as partners or carers). Most care decisions are therefore made in triadic interactions (conversations involving three parties): the PwND, healthcare professional (HCP) and the PwND's carer, meaning that decision making is a triadic process.

This study addresses the problem of how PwND can be more fully involved in decisions about their own care. NIHR-funded research [5] calls for better involvement of PwND in triadic decision making. While most PwND want to be involved in decision making about their own care [1,6], the involvement of carers in such processes can be helpful: for instance, they can clarify the history, and help ensure that decisions that have been made are recalled and executed. Carers can also support PwND in expressing their desires, values, and preferences even when they may lack decision-making capacity or the means of communicating their ideas clearly. Previous research on triadic medical encounters shows that the presence of a third party (in addition to the HCP and the patient) does not negatively affect the amount or content of patient contributions, nor the level of attention they receive from the HCP [7,8]. In fact, it has been shown how companion contributions can be beneficial, i.e. by introducing important information into the interaction, or supporting the patient [9,10]. However, in







relation to PwND it has also been demonstrated that carers may overlook the wishes of PwND and take over decision making [11], speaking for or about PwND rather than supporting them to speak for themselves [12]. This silencing of PwND can create problems for both the PwND and their carers when the experiences of PwND are distorted, and PwND are often treated as less competent than they are [13].

Furthermore, carers and PwND are not always aligned in their decision making around courses of treatment or other care decisions [11,14,15]: carers and PwND may differ in what they see as a 'reasonable decision' [16], and PwND may lack understanding of future changes to their condition [17]. This lack of ability to understand or accept the changes in their circumstances is one of the problems in involving PwND in shared decision making (SDM) [18]. As time passes, PwND may become unable to articulate personal information and caregivers may need to be helped effectively to engage in healthcare appointments on the PwND's behalf [15]. A detailed ethnographic study of triadic shared decision making in acute settings highlighted the difficulties in involving PwND in decision making due to the lack of timely and consistent engagement with the person by the care team, the effect of being taken out of their normal, known environment on the PwND's behaviour and presentation, and the general lack of knowledge about the PwND's understanding of their current situation [19; see also 20].

Legal and professional guidance on decision-making with PwND states that carers should only be involved insofar as they help HCPs understand the PwND's past and present views, or what would influence their decisions if they had capacity [2]. Existing legal frameworks thereby focus on protecting the interests of the patient and may not take full account of the consequences care-related decisions could have for others [21]. This is particularly relevant in neurodegenerative cognitive disorders, where carers might be directly affected by decisions about care provision (e.g. through refusal of formal carer support); or genetic testing (in inherited dementias). Our research will show how HCP can improve communication with PwND by taking optimal account of their views, and by involving carers when appropriate.

Existing NICE guidance on involving patients in decision making is not tailored for interactions involving PwND, and is based on research on two-party encounters [22,23] rather than the triadic decision making processes which are typical for healthcare encounters with PwND [4,24]. Prior work by this study's joint lead applicant has highlighted how much of a difference the presence of a third party makes to interactions between patients and HCPs [25]: Several groups of patients - in addition to PwND - are routinely accompanied to their appointments, for instance older adults without cognitive impairment [26], children [27], and adults with intellectual disabilities (ID) [28,29].

The similarities with interactions involving PwND are likely to be greatest with studies focusing on conversations involving adults with intellectual disabilities (ID), whose cognitive abilities are affected. When providing health information to adults with ID, HCPs need special skills in conducting three-way interactions [29], but little information about what these skills are or how they can be developed exists. Previous research exploring decision making and the deployment of choice in interactions involving adults with ID has characterised the difficulties that need to be overcome [29, 30]. These studies use







the method of Conversation Analysis (CA) to analyse how actual, real-life decision-making interactions are conducted.

A focus on actual medical interactions is increasingly advocated for understanding and improving healthcare [31,32], reflecting the mounting evidence that the precise wording used in a particular setting can make a big difference to the outcome of the conversation. A comparison of two closing questions used by GPs, "is there ANYthing else" vs."is there SOMEthing else you want to address today?" found that in response to "anything", patients were no more likely to express additional concerns than if the question was not asked at all. In contrast, "something" eliminated 78% of unvoiced concerns (odds ratio=.154, p=.001) [33].

Current work in this tradition includes two projects led by the co-PI: An RfPB-funded study (PB-PG-0211-24079) showed how conversational markers can support the differentiation of neurodegenerative and functional memory complaints. It took account of third party contributions but did not focus on decision-making [4,24]. An HSDR funded study (10/2000/61) demonstrated how HCPs can ensure patients know they have choices in clinic conversations [22] but did not focus on triadic environments or individuals with cognitive impairment. Our proposed research will use the same tried-and-tested methodology to build on these studies, and focus on decision-making involving PwND in a triadic medical environment.

CA has been successfully used to develop a variety of training packages, which will help underpin our public- and clinic-facing outputs. These range from training schemes designed to improve everyday communication (eg.,The Conversation Analytic Role-play Method (CARM [68]); to diagnostic aids (eg., for the diagnosis of epilepsy [69]; and for the differential diagnosis of functional and neurodegenerative memory disorders [4]); to therapeutic interventions aimed at (eg., improving communication between people with aphasia and their partners, [70]). The CA-based Better Conversations intervention training package [70] has been extended to various subtypes of aphasia, and the NIHR is currently funding another study extending the Better Conversations model to neurodegenerative cognitive disorders.

The outputs from this project will build on these successes which, like our study, all have a conversation analytic base. However, all the training and diagnostic packages described rely on the analysis of only the patient's speech, or of a dyadic interaction. This highlights the crucial difference between past work and our study - we are focussing on three-party conversations, which have a different underlying structure and a different set of turn-taking rules. The main aim of our first work package is to understand how triadic interactions differ from dyadic ones, and this basic understanding will then inform all the following work in the project.

2 RATIONALE

Dementia affects 944,000 people in the UK and is projected to rise to 1.1 million by 2030, and 1.4 million by 2040 [34]. As the disease progresses it becomes harder for PwND to communicate what they want and need, and ever more important to ensure an optimal transfer of information between PwND, carers and HCPs. Acknowledging public concerns and the continuing need for a better understanding of dementia, one of the most feared and stigmatised health conditions [35], the HS&DR has published a highlight notice for dementia to which this project is a direct response. Another NIHR-







funded study has recently highlighted the importance of better involvement of PwND in conversations about their care [5]. Our research will help with this by showing clinicians and carers how they can optimally involve PwND in important decisions about their own lives and thereby contribute to improving the health and wellbeing of both, PwND and their carers, two populations specifically identified as currently underserved by research [36].

PwND and their carers have to make many decisions about investigations and treatment related to their neurodegenerative cognitive disorders, and also about common comorbidities such as arthritis, hearing problems and heart disease [37]. As neurodegenerative cognitive disorders progresses, carers are likely to experience increasing pressure to get involved in decision-making or to take over decisions which the PwND would normally make themselves. This is at odds with the increasing emphasis on self-care, and on supporting people with neurodegenerative cognitive disorders to live independently as long as possible. Therefore our research is particularly timely as it will investigate how HCP can best involve PwND and their carers in such decisions, especially as the PwNDs' views may change as they become more disabled by their condition.

Our study will examine real life decision-making among HCP, PwND and their carers to find examples of best practice, and use these to develop training materials to enable PwND to better exercise their rights and participate in making choices about their care. People have the right to reach their own decisions regarding their health and care needs [1,2]. This right extends to all individuals with decision-making capacity [3]; for decisions where individuals lack capacity, those caring for them should make the decision in their overall best interests. Importantly, such best interests are determined not solely in terms of their medical interests, but with reference to their past and present wishes and feelings, and the beliefs and values that would have influenced them in coming to a decision had they had the capacity to make it [38].

While cognitive impairment can (though does not necessarily) affect the capacity of PwND, they may still be able to articulate authentic values that can be brought into the decision-making process; indeed, qualitative research involving PwND highlights that PwND can reliability report values and preferences even in moderate to severe stages of the condition [11]. Acknowledging the importance of supporting authentic decision-making – even in situations of impairment of capacity – has led to the rise of the person-centred care model for PwND [39, 40] operationalised in the shared decision-making (SDM) model [11,41]. While there is no universally-agreed definition for SDM [41], PwND interpret it as 'supporting' (rather than taking over) their participation in decision-making, 'hanging on' (versus letting go) to situations where they do maintain decision-making capacity, and 'being central to' (rather than being excluded from) making decisions [42].

Although we know that PwND perceive SDM positively [43-45], we still do not know just how SDM is, or should be, done. We lack concrete examples of what type of language use counts as 'supporting' participation; as 'hanging on' to situations where PwND retain capacity; as PwND 'being central to' rather than excluded from decision making. Instead, research to date mostly has relied on descriptions of SDM collected after the fact. Questions and topics such as those reported in [45] about what information was provided, what prompted the decision, what was the decision, did you reach it together, what were the alternatives are all excellent questions, but they require participants to recall events that have already happened. Instead, our study aims to capture what is said in decision-







making conversations and to analyse the language used. Our study will examine the detail of the language used to accomplish SDM (or show when it is not achieved), and this information can be used to provide concrete guidance for future encounters.

This particular gap in our understanding of the language used to implement SDM stems from the methodologies employed in previous research. Most studies of triadic medical interactions have taken a quantitative coding approach (eg. [26]), although the relevance of this is limited in decision-making, which is clearly sequential. Methodologies such as Conversation Analysis (CA), which focus on interactional aspects of discourse, are better-suited for studying how decisions evolve. CA is a qualitative, micro-analytic method for studying real-life interaction widely recognised as the leading methodology for investigating doctor-patient communication in practice [31,32]. Our own previous research has demonstrated how CA-derived insights can be the basis of an online learning resource judged to be capable of changing clinical practice and recommended by NICE [23,46].

Our proposed study will use this proven CA-methodology to investigate how clinicians design questions or present choices when there are two potential respondents, one of whom is a PwND. Explicit attention to the triadic nature of the consultations is important because the structure of conversation, in particular turn-taking, differs when more than two parties are involved. Most research to date involving PwND has concentrated on just two parties out of the three that are present in most encounters, eg., the 36 papers reviewed in [11], [17].

Researchers have begun to include PwND in the design and conduct of the research itself, not only as participants. PwND have been involved as participatory researchers to investigate ways in which a discussion tool about risk could be made more acceptable to PwND[18]; an ethnographic study involves PwND as equals alongside carers and HCPs [19]. Other NIHR-funded research [5] calls for greater involvement of PwND in triadic decision making. To address these goals, our study will engage a PwND and a carer as PPI co-applicants, and also contains a longitudinal component with 10 participant researcher dyads recording their interactions with healthcare professionals such as GPs and nurses.

3 THEORETICAL FRAMEWORK

The recognition of people's right to make their own decisions regarding their health and care needs [2,3] is grounded in the principle of respect for autonomy, which maintains that individual persons should be able to act freely in accordance with a self-chosen plan [47]. Three broad arguments are usually levelled in support of this principle [48]. One line of argument, owed to the utilitarian philosopher JS Mill, holds that permitting individuals to exercise free choice promotes the utility of all, by helping individuals to develop their capacities for reason, fulfil their own ideas of the good, and demonstrate 'experiments in living' that may display to others differing visions of individual and social welfare. Another draws from Immanuel Kant's categorical imperative, in its Formula of Humanity, that entreats moral agents to treat other humans, not merely as means, but as ends in themselves [49]; more precisely, according to Kant it is the human capacity to determine our own values and desires that give us all reason to act, and we ought therefore to treat others' values with the respect we show our own [50,51]. Lastly – least explicitly argued for, but comprising part of the 'background







constellation of values' widely accepted in contemporary medical ethics and law [52,53] – is the assumption of 'liberal neutrality', that there is intrinsic value in people being permitted to frame and pursue the good in their own way.

Understanding autonomy – and thus decision-making – in this fashion has important implications for the care of PwND. On all of these justifications of the principle of respect for autonomy, it is not simply the fact of being able to make decisions (decision-making capacity) that is important, but also that these decisions reflect what is most important to the person (decision-making authenticity).

Further philosophical insights into autonomy – and its implications for person-centred care and SDM – shape our framing of decision-making. The first is that autonomy is relational – social relationships can support or undermine autonomy [54]. Relational autonomy highlights that, for all people, our ability to articulate authentic values, exercise decision-making capacity, and express those choices, show interpersonal material, cognitive, and affective dependencies that can support or undermine these abilities [55,56,57]. For PwND, supportive social relationships can help them express treatment-related values, and to adhere to decisions made on the basis of those values, in the context of major healthcare decisions such as cancer treatment [58]. PwND identify the relationships that help them remain involved in SDM [11], and spouse-carers participate in SDM in ways that support [42] or 'manage' [59] the autonomy of PwND. However, such social relationships can also undermine the autonomy of PwND. A focus on capacity (over authenticity) can deprive PwND of the opportunity and support required to articulate their values [58,60]; carers can misidentify preferences [61] and can exclude PwND from decision-making when their only cognitive impairment is only mild [11].

A second core feature of respect for autonomy and SDM for PwND is that it is a temporally-extended process. While ethical dilemmas in healthcare are often framed in terms of the "punctate decision" – a specific act or choice, considered in isolation from the rest of a person's care [62] – the degree to which PwND can participate in these decisions depends upon: the ways in which other participants in SDM have attempted to understand PwND's preferences; the encouragement they have received to participate in SDM; or development of relationships that enhance or undermine PwND's confidence in their ability to express their values [63,64].

The third, related, consideration is that our authentic values may change over time, and indeed that neurodegenerative cognitive disorders itself may be a transformative experience [66]. Transformative experiences are those that not only change our authentic values, but do so in a way that is difficult to predict from before we have undergone the experience [66]. PwND may undergo just such a transition. As Wendy Mitchell, an author writing after a dementia diagnosis, put it in writing to her former self:

We wouldn't get on now, you and I. Too much time has passed. We are friends who have lost touch, who now lead parallel lives. We like different things. (Mitchell 2018)

This entails that expressed preferences of a PwND prior to the clinical onset of their condition do not necessarily express their present authentic preferences. SDM for PwND therefore requires attention to the person as they are now – and how that may differ from their former selves – rather than sole adherence to previously-expressed wishes (e.g. in the form of advance directives) [65].







This understanding of respect for autonomy – as relational, temporally-extended, and potentially transformative – underpins our framing of decision-making by and for PwND. We do not invoke a particular theoretical model of decision-making [67], since Conversation Analysis provides tools for a fine-grained analysis of different ways in which decisions are reached, and different approaches may be relevant in different contexts – and indeed, highlights that not all parties necessarily even agree on which points of a conversation represent 'choices' at all [22]. Instead, we aim to identify conversational and interactional practices that can support person-centred care, understood in terms of making decisions that reflect PwND's authentic values and preferences [39,40].

4 RESEARCH QUESTION/AIM(S)

The overall goal of the study is to analyse the strategies used by all parties in triadic interactions to either open up, or close down, opportunities for involvement of the PwND in decision making about their own care.

There are 4 key research aims:

- 1. To contribute to the evidence base about how decision making takes place in triadic environments
- 2. To compare findings on decision making in triadic medical encounters involving patients with and without neurodegenerative cognitive disorders
- 3. To identify the most effective and appropriate strategies for involving PwND in decisions about their medical care, especially as the disease progresses
- 4. To disseminate these findings to HCP, PwND and carers in an effort to increase patient involvement in decision-making about their care

And 6 study objectives:

- 1. Provide a detailed analysis of the communicative strategies used in triadic consultations and compare those involving patients with or without neurodegenerative cognitive disorders (Aim 1)
- 2. Identify how these strategies can support or limit shared decision-making for PwND (Aim 1, 2)
- 3. Use both cross-sectional and a novel longitudinal PwND/carer co-researcher design to describe how the meaning of shared decision-making for PwND evolves and communicative strategies used change as the disease progresses (Aim 2, 3)
- 4. Test the hypothesis that neurodegenerative cognitive disorders is a 'transformative experience', and the authentic values of PwND change as the disease progresses (Aim 3)
- 5. Construct educational resources and decision support tools to enable SDM for PwND (Aim 4)
- 6. Disseminate results to PwND and all involved in working with and caring for them (Aim 4)

5 STUDY DESIGN

This primary research study will analyse clinical decision-making involving PwND, their carers, and HCPs. It will use Conversation Analysis (CA) to analyse existing (WP1) as well as new cross-sectional (WP2) and longitudinal (WP3) datasets. CA is a qualitative, micro-analytic method for studying real-life







interaction widely recognised as the leading methodology for investigating doctor-patient communication in practice [31,32]. Our own previous research has demonstrated how CA-derived insights can be the basis of an online learning resource judged to be capable of changing clinical practice and recommended by NICE [23,46].

In WP 1, we will identify decision points in our existing audio and video recordings of previously transcribed triadic encounters involving patients without cognitive impairment [22]. These recordings were gathered in previous studies in which participants gave consent to use them in future studies. In addition we will draw on previous data from our RfPB study of diagnostic interactions in the memory clinic (including interactions involving participants with memory problems) [4]. Analysing the contributions of third parties in these datasets will allow us to identify differences in the role of the accompanying person when the patient has a neurodegenerative cognitive disorder and will enable us to gain initial insights into the effects of cognitive impairment on triadic decision-making.

In WP2, 100 clinical encounters between patients, carers, and HCP will be video and/or audio recorded in hospital- and community-based specialist memory clinics at both Sheffield Teaching Hospital NHS Foundation Trust, and the Rotherham Doncaster and South Humber NHS Foundation Trust (RDaSH). Apart from the recording of the interaction, our study will have no effect on the content or nature of these encounters. In our analysis of this cross-sectional dataset we will take account of the level of cognitive impairment of the participating patients. We will use the patient's standard cognitive score on either Addenbrooke's Cognitive Examination (ACE), Montreal Cognitive Assessment (MoCA) or Mini Mental State (MMSE) findings as a marker of the degree of cognitive impairment. These three tools are the most commonly used as the normal part of these consultations, and participants in WP2 of our study will be stratified on the basis of these scores.

WP3 is the longitudinal arm of the study. In this work package, 10 highly-motivated patient-carer dyads recruited from participants of the cross-sectional study will be involved as co-researchers, helping the research team to record further medical appointments in a range of medical settings (eg., with dementia nurses, occupational therapists, GPs) for the duration of our data collection period (up to two years).

The audio and video data collected in the study will be recorded onto encrypted SD cards, then transferred within 24 hours to secure storage on a University of Sheffield server. The data will be encrypted and shared via Google Drive with a professional transcription service who will decrypt the data, create the transcription, re-encrypt the data and return it through a Google drive folder. All identifying information (eg., proper names) will be anonymised in the transcripts. The video and audio data itself will not be anonymised but will be stored securely on a securely encrypted server at the University of Sheffield, and only members of the research team will have access to the data, unless participants specify otherwise on the consent form.

Participants will be asked to give explicit permission for the audio/video recordings to be used in future research, and for teaching purposes, and to allow short unidentifiable quotes from the transcript to be







included in publications. If participants specify that they do not want us to use their data for future research or for training purposes, it will be destroyed after 15 years. Data will then be erased from all computer storage facilities according to the standard university procedures in place at the time for deleting and handling confidential, digital data.

End of the Study

The end of the study will be the last analysis of video or audio recording of an included interaction.

6 STUDY SETTING

WP1 analyses existing data, collected in Sheffield outpatient neurology clinics as part of previous studies. All participants have previously provided consent to use this data in future research and therefore explicit consent for this work package will not be sought.

For WP2, participants will be recruited from referrals to hospital- and community-based specialist memory clinics at both Sheffield Teaching Hospitals NHS Foundation Trust, and the Rotherham Doncaster and South Humber NHS Foundation Trust (RDaSH). These are the only 2 sites involved in the study and the same type of activities are taking place at each site. There are no site specific requirements. Approximately 20 clinicians are involved in consultations across the two memory clinics, and all will be asked for their consent to be recorded.

For WP3, co-researcher participant dyads will be recruited from participants in WP2. They will work with the research team (primarily with the RA) to collect additional video recorded data over a period of two years in a variety of settings including community and primary care, eg., at GP appointments. During the funding application, we were advised by the HRA that this type of longitudinal data collection across a variety of settings was feasible. They stated: "the acute care Trust(s) where patient/carer consent is taken would be classed as a research site [eg., Sheffield Teaching Hospital NHS Foundation Trust, or the Rotherham Doncaster and South Humber NHS Foundation Trust (RDaSH).]. However, the GP practices and tertiary care sites would not be classed as a research site type as there was no requirement to process personal data and all activities would be as per standard of care. Therefore there is no need to inform relevant R&Ds or to gain 'generic' consent for sites. However, there is still the expectation that individual clinicians would consent to being recorded. Please note that the clinicians would not count as participants; their consent would purely be for the recording."

To maximize understanding of and participation in the study, an announcement of the study will be placed in a newsletter sent to Primary Care Sheffield (a GP consortium). Individualised letters will also be sent to the GP practices of the PwND taking part in WP3. Participants will be helped to produce an 'about me' card containing study information and contact details which they would introduce themselves to other healthcare professionals. Additionally, the RA will attend appointments with the







participant-researchers to answer questions and facilitate recording. The RA will also obtain consent at each appointment before the consultation commences and prior to recording.

7 SAMPLE AND RECRUITMENT

7.1 Eligibility Criteria

For WP2, only patients who bring a companion/carer with them to their appointment will be eligible. All patients meeting the inclusion criteria will be invited to take part in the study. Some will be follow-up appointments, but some will be attending the memory clinic for the first time. Not all of these new patients will receive a diagnosis of a neurodegenerative cognitive disorder, and if they do not, their data will not be used in this study. Both the patient and their companion/carer must agree to participate or they cannot take part in the study. We will continue recruiting participants until we have recorded 100 consultations with people who already have a diagnosis of neurodegenerative cognitive disorder as a result of this consultation. Only data from people with a neurodegenerative cognitive disorder diagnosis will be analysed for this study, and only people with a neurodegenerative cognitive disorder diagnosis will be approached for participation in WP3.

Patients who consent to take part in data collection for WP2 will be asked for consent to contact them about future studies. If they consent to this, and if they receive a diagnosis of neurodegenerative cognitive disorder, they will be eligible for WP3, the longitudinal study. Recruitment for WP3 will begin simultaneously with WP2 so we do not expect any difficulty in recruiting 10 dyads in a timely fashion. Therefore the inclusion and exclusion criteria for participants in WP3 are the same as WP2.

7.1.1 Inclusion criteria

- Able to give informed consent
- Two people attending the appointment, in a carer/cared for relationship.
- English speakers, whether as a first or other language.
- Only participants who have already received or are receiving a diagnosis of neurodegenerative cognitive disorder will have their recorded data analysed for this study.

7.1.2 Exclusion criteria

- Unable to give informed consent
- Attending the appointment singly or with someone that is not part of an ongoing caring relationship
- Unable to read/write/understand English

7.2 Sampling







For WP2, we aim to collect 100 consultations (50 per site). The chosen methodology, Conversation Analysis, is an intensive, detail-oriented approach and this number of consultations strikes the appropriate balance between providing enough different interactions and being pragmatic about the time necessary for transcription and analysis of the data. We have allowed for 7 months of data collection at STH and RDASH to ensure we collect enough data to analyse cross-sectionally within the time constraints of the study, and also to simultaneously recruit participants for the longitudinal study (WP3). We will recruit both follow-up and new referrals to the memory clinics, as for new referrals only 60% of the consultations will result in a neurodegenerative cognitive disorders diagnosis. We will include PISs in all memory clinic appointment letters, and consent and record all who indicate an interest in taking part. We will remove the data from the analysis, however, if they are not diagnosed with a neurodegenerative disorder.

For WP3, we will recruit a purposive sample with the aim of achieving maximum variation across categories theoretically predicted to influence shared decision-making (eg., race, age, gender, education level, type of kinship/carer relationship). Race influences HCPs' interactions with patients, and may intersect with other inequities in ways that specifically undermine the capacities of PwND from racialised and otherwise marginalised communities to express their authentic care-related preferences. Supporting shared decision-making for diverse communities will require understanding these intersections, therefore we aim to recruit a sample varying across race, gender, age, education level and type of kinship/carer relationship (this information will have been collected as part of WP2).

We aim to recruit 10 co-researcher participant dyads over a period of up to 2 years. Again, we are taking an approach which balances spread across different environments against time to process and analyse the data alongside the likely changes in capacity of the co-researcher participants. Previous audits suggest that patients will receive 1-3 follow-up appointments per year in the specialist memory services; they are also likely to have appointments elsewhere (e.g. in primary care). Feedback from PPI groups indicated that we should also allow the co-researcher participant pairs to suggest settings they would find beneficial for research, such as home visits from occupational therapy, or even dentist/optometrist appointments, so we are confident of our ability to record numerous encounters with 10 participant pairs.

7.3 Recruitment

WP2: Participants will be people with memory complaints and their care partners attending the memory clinics at the Sheffield Teaching Hospital and RDaSH. Potential participants will receive a short PIS about the study with their appointment letter. The short PIS will contain basic information about the study as well as a phone number, email address and website address to contact for more information. It will also inform them that a member of the research team will be present before their clinic appointment to discuss the study and answer any questions they may have.







When potential participants attend their appointment, receptionists (i.e. members of the clinical care team) will ask patients and companions whether they are interested in taking part in our study, and if they would like to talk to the clinical trial assistant/other member of the research team present in the clinic. This member of the research team would then explain the study again and record people's consent, providing them at this time with the longer PIS. This method of providing information about the study and taking consent was suggested by the PPI co-applicants on this project. They believe that the best way to recruit participants and to make sure they understood the study information was through conversation, rather than written PIS. A full PIS document will be given to the participants once the research team is satisfied that they understand the study, to comply with all necessary legal and ethical guidance

The participants will be informed that taking part in the study will not affect their care or the time it takes to get their diagnosis. They will also be told that they can stop the recording at any time during or at the end of the appointment, and that they can withdraw their data from the study for up to 2 weeks after their consultation and all recordings and data collected as part of the study will be destroyed. After consent is taken, the couple will each be asked to fill out a short form capturing demographic information: DOB, gender, carer relationship (eg., spouses, parent-child) and ethnic group membership.

A recruitment log will be maintained by the local team at each site which will assign each recruited participant a trial ID number. This will be kept separately to the trial data in a secure location on NHS premises (or NHS servers if electronic). Any data collected for the study which can be pseudonymised using this trial ID (e.g. excluding video and audio data) will be stored in a pseudonymised format.

The consultation room will have a video camera set up, which will be switched on by a member of the research team (who will then leave the room) or by the clinician themselves. The consultation will then proceed as normal. During the consultation the clinician will administer the Addenbrooke's Cognitive Examination (ACE), the standard cognitive screening test used in both clinics, and the results of this will be shared with the research team to utilise in the study. Scores on the ACE will be uploaded to a password-protected electronic database stored on the University of Sheffield's servers. This database will contain no participant identifiable data and link the data to participants via their pseudonymised trial ID.

WP3: Participants in WP2 will have indicated on their original consent form whether they consent to be contacted about additional studies. Patients receiving a diagnosis of neurodegenerative memory disorder as identified by one of the consultant neurologists working in the STH memory clinic after their consultation recorded in WP2 will be invited to take part in the longitudinal study. Using information from the demographic survey, we will recruit participants from as diverse a range of PwND and carers as possible, varying across race, gender, and age. Potential participants will be contacted as per their preference to discuss taking part in the longitudinal study. After this conversation, with their permission, they will be sent a written PIS detailing WP3. Within one week, the RA will again contact the pair via their preferred method to answer any questions and take consent. Once we reach a total of 10 co-researcher participant pairs we will stop recruitment.

7.3.1 Consent







Some participants recruited into the study at WP2 will not yet have a diagnosis of neurodegenerative cognitive disorder as we will be recruiting them at their first point of contact with the memory clinics, therefore we have no reason to believe either member of the pair would be lacking capacity. However, recruitment and consent will be carried out by research team members trained in mental capacity assessments, who will be able to assess their ability to make a free choice, to understand the research, and to be capable of making a decision. Participants will receive study introduction flyers (short PIS) before attending their appointment, and at their appointment will be given the chance to ask questions and discuss the study with members of the research team including the clinician they are seeing. Once the research team member doing the consenting is satisfied that they understand the research and wish to take part, they will be provided with written participant information sheets and consent forms which have been co–produced with the PPI co-applicants working in the study team, and approved by the REC.

Participants recruited as part of WP3 will join the study team as co-researcher participants. Members of the research team with specialist training in assessing mental capacity will be in close contact (eg., on a fortnightly or monthly basis) with these participant-researchers. We will ask the member of the participant-research pair who has a neurodegenerative cognitive disorder to agree to continue taking part even if their capacity fluctuates, and if they lose the capacity to consent. This will not be treated as an absolute advance directive, however. We will ask participants to nominate a consultee at the start of the study (who is a different person from the participant-carer). If it is deemed that the PwND at any point no longer has capacity to continue to give consent to participation, we will ask the consultee to act in determining their best interests, keeping in mind that the PwND had previously agreed to continue in the study but balancing this with their ability to communicate or any agitation being caused.

The RA will contact the nominated personal consultees and send them an information sheet about the study and what being a consultee entails. Consultees will have up to 2 weeks to decide whether or not to take on the role, and will be asked to sign a written declaration form.

8 ASSESSMENT AND MANAGEMENT OF RISK

Our sponsor (Sheffield Teaching Hospitals NHS Trust) has advised that they are satisfied that this study is not performing an intervention, nor conducting assessments which offer any potential risk to the participants. Given the nature of neurodegenerative cognitive disorders, the main issue we may face is a loss of capacity to consent to continue taking part. Our procedure for dealing with this is outlined in section 7.3.1 of this protocol.

The study will comply with the Research Governance procedures required by the sponsor, and will be conducted in accordance with the principles of the Declaration of Helsinki.

Transcripts of the recordings will be pseudonymised, with all proper names, places and any other identifiable information replaced. None of the extracts that appear in our publications will allow the identification of individual participants. Additionally, video or audio recordings will only be used for teaching or research purposes if participants have explicitly consented for this.







9 PATIENT & PUBLIC INVOLVEMENT

This project has 4 co-applicant PPI members. They have been/are involved in assessing the acceptability of the research, management of the research, and dissemination of findings. They have provided input on the PIS and consent forms, and meet with the research team on a regular (quarterly) basis to review the study. We also have a member of the public on the Study Steering Committee.

The PPI members were keen that we think beyond adopting the traditional approach to providing lengthy written PIS for prospective participants to read on their own. They suggested that a personal, conversational approach would be the best way of ensuring that potential participants truly understood the research they were being asked to take part in, noting that they themselves have failed to thoroughly read some PIS they have been presented with. Once we explained that we cannot contact potential participants without their consent, they were satisfied that a short PIS introducing the study, sent through the post, followed up by face to face conversation in the memory clinics, would encourage people to take part and ensure that they had a better understanding of the research than being given a written PIS to read and sign, even if they are encouraged to ask questions about it. However we are aware of the legal and ethical importance of providing a written document, and will present this to participants once they have spoken with a member of the research team, shown they understand the research, and indicated their willingness to take part. Both short and long versions of the PIS have been vetted by the PPI members.

10 FUNDING AND INDEMNITY ARRANGEMENTS

The study and its activities will be covered by NHS indemnity. £493,292 of funding has been received to carry out this project from the NIHR Health and Social Care Delivery Research (HSDR) Dementia Call 21/505.

11 DISSEMINATION AND TAKING THE WORK FORWARD

The overall dissemination strategy will be facilitated by the team's close links with professional organisations such as the Royal College of Physicians, Association of British Neurologists, Alzheimer Research UK and General Practice. We have enlisted a study collaborator, Dr Danielle Jones (University of Bradford), who has extensive experience in developing educational materials for clinical and non-clinical health care workers. Her expertise will assist us in building a relationship with Health Education England, to offer summaries of our findings for inclusion in their E-learning modules. The Higher Education Dementia Network (HEDN) launches a new strategy in May, and we will explore this route for dissemination to HEI's, to seek ways to embed our findings within training and education for HCP students on a national level.

Specifically, findings will be disseminated in the following ways:

Dissemination to healthcare professionals and policy makers: the main outputs of the study will be detailed guidance for carers and HCP about (i) how best to facilitate shared decision making and (ii) what pitfalls to avoid. One of the co-investigators chairs the Association of British Neurologists







Specialist Interest Group in Cognitive Impairment and sits on the Dementia Translational research Centre (https://www.nihr.ac.uk/explore-nihr/support/dementia.htm) and thus can share information from this project locally, regionally, and nationally. To ensure that GPs are aware of our work, one of the co-investigators will lead on targeting GPs in Sheffield via their federation (Primary Care Sheffield), in the weekly primary care seminar at the University of Sheffield, via the North regional meeting of the Society of Academic Primary Care (SAPC), the national SAPC annual meeting, and the Royal College of GPs (RCGP). We will work with our local Health Care Partnership in South Yorkshire and Bassetlaw to disseminate within the NHS and care system. Finally, we will host a half-day national workshop to engage with clinicians and policy-makers, review the study results and present the implications and guidelines. The PPI co-applicants will be involved in preparing and delivering this workshop.

Dissemination to academic researchers: the study results will be submitted for publication within peer reviewed health communication and healthcare journals, and presented at appropriate international healthcare and communication conferences. In addition, the study and results will be presented to academic researchers and healthcare professionals through relevant seminars, including the Health Sciences School seminar series at the University of Sheffield, and at the Grand Round for medical members of staff at Sheffield Teaching Hospitals NHS Foundation trust.

Dissemination to people with dementia and carers: in addition to the provision of a full report, technical appendices and executive summary, a lay summary of the study's findings will be prepared with input from the PPI co-applicants and made available in written and short video format to key patient websites and newsletters. We will also seek to present our findings at local dementia advocacy group meetings, eg., quarterly SYDEMRAG meetings. Leaflets summarising the study findings and directing interested parties to further information (eg., charity websites as well as our own self-learning package) will be made available to GPs and other community healthcare settings, including dissemination to existing, marginalised patient groups such as Israac Somali and a South Asian community group that currently collaborate with research team members on another project. Research Champions have been trained from the Israac community group and we have created publicity films and posters to be shared more widely with Israac and other ethnic minority groups. Furthermore, within the workshops we run on what neurodegenerative cognitive disorder is and how it is diagnosed, we will include groups on decision making, as it is particularly important for people from more marginalised groups to understand they have the right to be included. We will also disseminate results through our DeepEnd GP practices which cover parts of Sheffield with higher levels of deprivation (https://www.yorkshiredeependgp.org/).

We will use our findings to develop guidance that will be packaged in the form of two online selfdirected learning platforms. Separate platforms will be developed for HCPs and carers. These learning platforms will involve guided analyses of video-samples of successful (appropriately inclusive) and less successful (exclusive) communication strategies for triadic decision making with PwND.

Additionally, we will produce short videos and leaflets that outline the study findings for carers and PwND. These will be designed with the input and oversight of the PPI co-applicants and the







participant-researcher dyads to ensure a high level of authenticity and acceptability. Providing both these types of public-facing outputs will ensure maximum outreach regardless of either digital or written literacy, whilst also providing information in forms that are accessible and acceptable to people of all ages. Written information will also be readily translatable for further outreach to non-English speaking communities.

12 REFERENCES

- 1. Department of Health and Social Care. *G8 Dementia Summit Communique* December 2013. <u>https://www.gov.uk/government/publications/g8-dementia-summit-agreements/g8-dementia-summit-communique</u>
- 2. General Medical Council Decision Making and Consent Guidance. <u>https://www.gmc-uk.org/ethical-guidance/ethical-guidance-for-doctors/decision-making-and-consent</u>
- 3. Department of Health and Social Care. *NHS Constitution for England*, last updated 1 January 2021. <u>https://www.gov.uk/government/publications/the-nhs-constitution-for-england</u>
- Reuber M et al. An interactional profile to assist the differential diagnosis of neurodegenerative and functional memory disorders. *Alzheimer Dis Assoc Disord* 2018;32(3):197-206. doi: 10.1097/ WAD.00000000000231.
- 5. <u>https://evidence.nihr.ac.uk/alert/people-dementia-more-involvement-long-term-care-decisions/</u>
- Hirschman KB, Joyce CM, James BD, Xie SX, Karlawish JHT. Do Alzheimer's disease patients want to participate in a treatment decision, and would their caregivers let them? *Gerontologist*. 2005;45(3):381–8. doi:10.1093/geront/45.3.381
- Shields CG, Epstein RM, Fiscella K, Franks P, McCann R, McCormick K, et al. Influence of accompanied encounters on patient-centeredness with older patients. *J Am Board Fam Pract*. 2005;18(5):344–54. doi:10.3122/jabfm.18.5.344
- Ishikawa H, Roter DL, Yamazaki Y, Takayama T. Physician-elderly patient-companion communication and roles of companions in Japanese geriatric encounters. *Soc Sci Med*. 2005;60(10):2307–20. doi:10.1016/j.socscimed.2004.08.071
- 9. Street RL, Gordon HS. Companion participation in cancer consultations. *Psychooncology*. 2008;17(3):244–51. doi:10.1002/pon.1225
- Clayman ML, Roter D, Wissow LS, Bandeen-Roche K. Autonomy-related behaviors of patient companions and their effect on decision-making activity in geriatric primary care visits. *Soc Sci Med*. 2005;60(7):1583–91. doi:10.1016/j.socscimed.2004.08.004
- 11. Miller LM, Whitlatch CJ, Lyons KS. Shared decision-making in dementia: A review of patient and family carer involvement. *Dementia* 2016;15(5):1141–1157. doi:10.1177/1471301214555542
- 12. Nilsson E, Ekström A, Majlesi AR. Speaking for and about a spouse with dementia: A matter of inclusion or exclusion? *Discourse Studies* 2018; 20(6):770–791. doi:10.1177/1461445618770482
- 13. Österholm JH, Samuelsson C. Orally positioning persons with dementia in assessment meetings. *Ageing Soc* 2015;35(2):367–88. doi:10.1017/S0144686X13000755
- Fried TR, Bradley EH, Towle VR. Valuing the outcomes of treatment: do patients and their caregivers agree? Arch Intern Med 2003; 163(17): 2073–78. : https://www.alzheimers.org.uk/sites/default/files/2019-11/cpec report november 2019.pdf







- Griffin JM, Riffin C, Havyer RD, Biggar VS, Comer M, Frangiosa TL, et al. Integrating Family Caregivers of People With Alzheimer's Disease and Dementias into Clinical Appointments: Identifying Potential Best Practices. J Appl Gerontol 2020;39(11):1184–94. doi:10.1177/0733464819880449
- 16. Horton-Deutsch S, Twigg P, Evans R. Health care decision-making of persons with dementia. *Dementia*. 2007;6(1):105–20. doi:10.1177/1471301207075643
- 17. Dening KH, King M, Jones L, Sampson EL. Healthcare decision-making: Past present and future, in light of a diagnosis of dementia. *Int J Palliat Nurs* 2017;23(1):4–11. doi:10.12968/ijpn.2017.23.1.4
- Goeman DP, Dickins M, Iliffe S, Pond D, O'Keefe F. Development of a discussion tool to enable wellbeing by providing choices for people with dementia: A qualitative study incorporating codesign and participatory action research. *BMJ Open* 2017;7(11). doi:10.1136/bmjopen-2017-017672
- 19. Kelley R, Godfrey M, Young J. Knowledge exchanges and decision-making within hospital dementia care triads: An ethnographic study. Gerontologist. 2021;61(6):954–64. doi:10.1093/geront/gnaa216
- Griffiths S, Gude A, Greene L, et al. 'Do I have the capacity to make capacity judgements?' Researcher reflections from a person-centred dementia support study. *Dementia*. 2022;2. doi:<u>10.1177/14713012211067320</u>
- 21. Herring J. Caregivers in medical law and ethics. *J Contemp Health Law and Policy* 2009; 25(1):1–37.
- 22. Reuber M, Toerien M, Shaw R, Duncan R. Delivering patient choice in clinical practice: a conversation analytic study of communication practices used in neurology clinics to involve patients in decision-making. *Health Serv Deliv Res* 2015;3(7).
- 23. NICE Guidelines. Shared Decision Making 2021 https://www.nice.org.uk/guidance/ng197
- Jones D, et al. Conversational assessment in memory clinic encounters: interactional profiling for differentiating dementia from functional memory disorders, *Aging Ment Health* 2016; 20(5): 500-509 DOI: <u>10.1080/13607863.2015.1021753</u>
- 25. Robson C, Drew P, Reuber M. Duration and structure of unaccompanied (dyadic) and accompanied (triadic) initial outpatient consultations in a specialised seizure clinic. *Epilepsy Behav* 2013; 449-454 doi.org/10.1016/j.yebeh.2013.03.008
- 26. Wolff JL et al. Examining the context and helpfulness of family companion contributions to older adults' primary care visits. *Patient Educ Couns* 2017; 100(3): 487-494. doi.org/10.1016/j.pec.2016.10.022
- 27. Stivers T. Managing patient pressure to prescribe antibiotics in the clinic. *Pediatr Drugs*. 2021;23(5):437–43. doi:10.1007/s40272-021-00466-y
- Antaki C, Chinn D. Companions' dilemma of intervention when they mediate between patients with intellectual disabilities and health staff. *Patient Educ Couns*. 2019;102(11):2024–30. doi:10.1016/j.pec.2019.05.020
- 29. Mastebroek M, Naaldenberg J, Lagro-Janssen AL, van Schrojenstein Lantman de Valk H. Health information exchange in general practice care for people with intellectual disabilities-A qualitative review of the literature. *Res Dev Dis* 2014;35(9):1978–87. doi:10.1016/j.ridd.2014.04.029
- Antaki C, Finlay WML, Walton C. Choices for people with intellectual disabilities: Official discourse and everyday practice. *J Policy Pract Intellect Disabil* 2009;6(4):260–6. doi:10.1111/j.1741-1130.2009.00230.x
- Drew P, Chatwin J, Collins S. Conversation analysis: a method for research into interactions between patients and health-care professionals. *Health Expect* 2001;4:58–70. http://dx.doi.org/ 10.1046/j.1369-6513.2001.00125.x
- 32. Heritage J, Maynard DW. *Communication in Medical Care: Interactions Between Primary Care Physicians and Patients*. Cambridge: Cambridge University Press; 2006.







- 33. Heritage J, Robinson JD, Elliott MN, Beckett M, Wilkes M. Reducing patients' unmet concerns in primary care: The difference one word can make. *J Gen Intern Med* 2007;22(10):1429-33.
- 34. [https://www.dementiastatistics.org/statistics/prevalence-projections-in-the-uk-2/]
- 35. Alzheimers Research UK, Dementia Attitudes Monitor 2019. https://www.dementiastatistics.org/statistics/perceptions-of-dementia/
- 36. https://www.nihr.ac.uk/documents/nihr-highlight-notice-dementia/27316
- 37. Prime Minister's Challenge on Dementia 2020. assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/414344/p m-dementia2020.pdf
- 38. Mental Capacity Act 2005. https://www.legislation.gov.uk/ukpga/2005/9/contents
- 39. McCormack B. Person-centredness in gerontological nursing: an overview of the literature. *J Clin Nurs* 2004;13(s1): 31–38. https://doi.org/10.1111/j.1365-2702.2004.00924.x
- 40. Edvardsson D, Winblad B, Sandman PO. Person-centred care of people with severe Alzheimer's Disease: Current status and ways forward. *The Lancet Neuro* 2008;7(4):362–67. https://doi.org/10.1016/S1474-4422(08)70063-2
- 41. Daly RL, Bunn F, Goodman C. Shared decision-making for people living with dementia in extended care settings: a systematic review. *BMJ Open* 2018;8(6): e018977. https://doi.org/10.1136/bmjopen-2017-018977.
- 42. Fetherstonhaugh D, Rayner JA, Tarzia L. Hanging on to some autonomy in decision making: how do spouse carers support this? *Dementia* 2019;18(4):1219–36. <u>https://doi.org/10.1177/1471301216678104</u>.
- Lawani MA, Turgeon Y, Côté L, Légaré F, Witteman HO, Morin M, et al. User-centered and theorybased design of a professional training program on shared decision-making with older adults living with neurocognitive disorders: a mixed-methods study. *BMC Med Inform Decis Mak* 2021;21(1). doi:10.1186/s12911-021-01396-y
- 44. Groen-Van De Ven L, Smits C, De Graaff F, Span M, Eefsting J, Jukema J, et al. Involvement of people with dementia in making decisions about their lives: A qualitative study that appraises shared decision-making concerning daycare. *BMJ Open* 2017;7(11). doi:10.1136/bmjopen-2017-018337
- Groen van de Ven L, Smits C, Elwyn G, Span M, Jukema J, Eefsting J, et al. Recognizing decision needs: first step for collaborative deliberation in dementia care networks. *Patient Educ Couns* 2017;100(7):1329–37. doi:10.1016/j.pec.2017.01.024
- 46. <u>https://www.nice.org.uk/guidance/ng197/resources/shared-decision-making-learning-package-9142488109</u>
- 47. Beauchamp TL, Childress JF. *Principles of Biomedical Ethics*. 8th edition. Oxford: Oxford University Press; 2019.
- 48. Wardrope A. Autonomy as ideology: towards an autonomy worthy of respect. *The New Bioethics* 2015b;21(1):56–70. https://doi.org/10.1179/2050287715Z.0000000057.
- 49. Kant I. *Groundwork of the Metaphysics of Morals*. C Korsgaard (ed). 2nd edition. Cambridge: Cambridge University Press; 2012.
- 50. Korsgaard C. *The Sources of Normativity*. Cambridge, New York: Cambridge University Press; 1996b.
- 51. Korsgaard C. *Creating the Kingdom of Ends*. Cambridge: Cambridge University Press; 1996a. https://doi.org/10.1017/CBO9781139174503.







- 52. Callahan D. Individual good and common good: a communitarian approach to bioethics. *Perspec Biol Med* 2003;46(4): 496–507. https://doi.org/10.1353/pbm.2003.0083.
- 53. Dawson A. The future of bioethics: three dogmas and a cup of hemlock. *Bioethics* 2010;24 (5): 218–25. https://doi.org/10.1111/j.1467-8519.2010.01814.x.
- 54. Mackenzie C, Stoljar N (eds.). *Relational Autonomy: Feminist Perspectives on Autonomy, Agency, and the Social Self.* New York: Oxford University Press; 2000.
- 55. Mackenzie C. Relational autonomy, normative authority and perfectionism. *J Soc Phil* 2008;39 (4): 512–33. https://doi.org/10.1111/j.1467-9833.2008.00440.x.
- 56. Colburn B. Autonomy and adaptive preferences. *Utilitas* 2011;23(1): 52–71. https://doi.org/10.1017/S0953820810000440.
- 57. Wardrope A. Liberal individualism, relational autonomy, and the social dimension of respect. *IJFAB: Int J Fem Approach Bioeth* 2015a;8 (1): 37–66.
- 58. Cook PS, McCarthy AL. Cancer treatment decision-making with/ for older adults with dementia: the intersections of autonomy, capital, and power. *Health Soc Rev* 2018;27(2): 184–98. https://doi.org/10.1080/14461242.2018.1466187.
- 59. Berry B, Apesoa-Varano EC, Gomez Y. How family members manage risk around functional decline: the autonomy management process in households facing dementia. *Soc Sci Med* 2015; *1982*(130): 107–114. https://doi.org/10.1016/j.socscimed.2015.02.014.
- 60. Harding R. Legal constructions of dementia: discourses of autonomy at the margins of capacity. *J* Soc Welf Fam Law 2012;34 (4): 425–42. https://doi.org/10.1080/09649069.2012.755031.
- 61. Whitlatch CJ, Piiparinen R, Friss Feinberg L. How well do family caregivers know their relatives' care values and preferences? *Dementia* 2009;8(2):223–43. https://doi.org/10.1177/1471301209103259.
- 62. Kukla R. Conscientious autonomy: displacing decisions in health care. *Hastings Center Report* 2005; 35 (2): 34–44. https://doi.org/10.1353/hcr.2005.0025.
- 63. Smebye KL, Kirkevold M, Engedal K. How do persons with dementia participate in decision making related to health and daily care? a multi-case study. *BMC Health Serv Res* 2012;12(1):241. https://doi.org/10.1186/1472-6963-12-241.
- 64. Sinclair C, et al. "A Real Bucket of Worms": Views of People Living with Dementia and Family Members on Supported Decision-Making'. *J Bioeth Ing* 2019;16(4):587–608. https://doi.org/10.1007/s11673-019-09945-x.
- 65. Walsh E. Cognitive transformation, dementia, and the moral weight of advance directives. *Amer J Bioethics* 2020;20(8):54–64. https://doi.org/10.1080/15265161.2020.1781955
- 66. Paul LA. *Transformative Experience*. Oxford: Oxford University Press; 2014.
- Bekker H. Using decision making theory to inform clinical practice in G Elwyn, A Edwards, R Thompson (eds) *Shared Decision Making: Evidence Based Patient Choice in Healthcare*, pp 45-52; Oxford: Oxford University Press; 2009.
- 68. Stokoe, E. (2014). The Conversation Analytic Role-play Method (CARM) A method for training communication skills as an alternative to simulated role-play. *Research on Language and Social Interaction*, 47 (3), 255-265, DOI: <u>10.1080/08351813.2014.925663</u>
- 69. Jenkins, L. & Reuber, M. (2014). A Conversation Analytic Intervention to Help Neurologists Identify Diagnostically Relevant Linguistic Features in Seizure Patients' Talk, *Research on Language and Social Interaction*, 47:3, 266-279, DOI: 10.1080/08351813.2014.925664







 Best, W., Maxim, J., Heilemann, C., Beckley, F., Johnson, F, Edwards, S.I., Howard, D., Beeke, S. (2016). Conversation Therapy with People with Aphasia and Conversation Partners using Video Feedback: A Group and Case Series Investigation of Changes in Interaction. *Frontiers In Human Neuroscience* 10(562) doi: 10.3389/fnhum.2076.00562